## **Research Paper**



# The Effect of Core Stability Exercises Combined With Abdominal Hollowing on Postural Balance in Patients With Non-specific Chronic Low Back Pain: A Randomized Controlled Trial

Ali Yalfani 1\* 💿 , Mahsa Mohamad Khani 1 💿 , Mohamadreza Ahmadi 1 💿 , Azadeh Asgarpoor 1 💿

1. Department of Corrective Exercises and Sport Injury, Faculty of Physical Education and Sport Sciences, Bu-Ali Sina University, Hamedan, Iran.



**Citation** Yalfani A, Mohamad Khani M, Ahmadi M, Asgarpoor A. The Effect of Core Stability Exercises Combined With Abdominal Hollowing on Postural Balance in Patients With Non-specific Chronic Low Back Pain: A Randomized Controlled Trial. Physical Treatments. 2023; 13(3):165-174. http://dx.doi.org/10.32598/ptj.13.3.442.5

doj<sup>\*</sup>: http://dx.doi.org/10.32598/ptj.13.3.442.5

## 

Article info: Received: 31 May 2023 Accepted: 15 Jun 2023 Available Online: 01 Jul 2023

#### **Keywords:**

Low back pain, Postural balance, Hollowing, Exercise

## ABSTRACT

**Purpose:** Altered movement patterns and spine instability are crucial factors in the etiology of chronic non-specific low back pain (CNSLBP). The faulty recruitment pattern of core muscles causes poor support and increased loading on the lumbar spine, ultimately leading to postural instability. This study aims to examine the effect of core stability exercises (CSE) combined with abdominal hollowing (AH) on postural balance in patients with CNSLBP.

**Methods:** The present study is a double-blind clinical trial. Thirty women with CNSLBP were recruited for the clinical trial. Patients were randomly assigned to experimental and control groups (15 patients in each group). The experimental group followed CSE with AH for 8 weeks. The control group received no intervention. Primary outcomes included pain (10 cm visual analogue scale), disability (Oswestry questionnaire), and proprioception (goniometer). Secondary outcomes included static and dynamic balance (biodex balance system). Analysis of covariance (ANCOVA) was used to compare the results and confidence intervals between groups.

**Results:** The experimental group compared to the control group with a high effect size had a significant difference in reducing pain (P=0.021,  $\eta p^2$ =0.183), disability (P=0.007,  $\eta p^2$ =0.237), and postural reconstruction error (P=0.000,  $\eta p^2$ =0.566). Also, the experimental group compared to the control group with a high effect size had a significant difference in the improvement of static (P=0.008,  $\eta p^2$ =0.367) and dynamic balance (P=0.008,  $\eta p^2$ =0.757).

**Conclusion:** The CSE combination with AH affects three active, passive, and nervous systems. It seems that the CSE combination with AH can modify the faulty movement control in local muscles, proprioception restoration, and subsequently lead to postural stability.

\* Corresponding Author:

Ali Yalfani, Professor.

Address: Department of Corrective Exercises and Sport Injury, Faculty of Physical Education and Sport Sciences, Bu-Ali Sina University, Hamedan, Iran. Phone: +98 (918) 3155478 E-mail: yalfani@basu.ac.ir

## Highlights

- Altered movement patterns and spinal instability are the cause of low back pain.
- Core local muscle co-contraction is necessary for spine stability.
- The stability exercises increase local muscle-isolated activation.
- The local muscle co-contraction leads to postural stability.

## Plain Language Summary

Patients with chronic non-specific low back pain (CNSLBP) show weakness of core muscles and faulty lumbarpelvic movement patterns. Weakness of stabilizing and respiratory muscles leads to instability and loading of the spine and ultimately low back pain. It seems that the combination of core stability exercises and abdominal hollowing by core muscle activation increases spine stability and postural balance in patients with back pain.

## 1. Introduction

ow back pain (LBP) is a painful discomfort defined between the costal margin and the gluteal folds, with or without referred pain in the legs [1]. A systematic review, including 54 countries reported that the prevalence of LBP in the population and

lifetime is 18% and 40%, respectively [2]. The healthcare cost of LBP is estimated to be £14 billion in the United Kingdom and \$14.5 billion in the United States [3]. In addition, LBP affects work absence and daily activities and causes depression, anxiety, low quality of life, and disability [4]. LBP is classified into two categories: Non-specific and mechanical [4]. Chronic non-specific low back pain (CNSLBP) accounts for 90%-95% of low back pain and affects 20% of the world's population [5, 6]. CNSLBP is a complex condition with unknown pathophysiology that occurs for three months or more [1, 7].

Although the cause of CNSLBP is still unclear, altered movement patterns and spinal instability seem to be crucial factors [8]. The spine's stability consists of three subsystems: Passive (bones, ligaments, and joint capsule), active (muscles and tendons), and nervous (central, and peripheral nervous system). According to the Punjabi model, the interaction of subsystems leads to movement control and spine stability [9]. In the nervous subgroup, proprioception is the main component of the sensory-motor system and the muscle spindle is the vital sensory source [10, 11]. Sensory information is processed in the spinal cord, brain stem, cerebral cortex, and cerebellum, which is vital in understanding the biomechanical and spatial position of the segments [10]. A previous study reported that proprioception is faulty in patients with CNSLBP [12]. Reduced proprioception feedback from mechanoreceptors at the pain location affects postural control [13]. Proprioception deficits in patients with LBP may reduce the maintenance of neutral spine posture and muscle coordination [11]. Muscle cocontraction is necessary to maintain the mechanical stability of the spine [14]. In healthy, the transverse abdominis (TrA) and multifidus (MF) muscles are activated as they feedforward before movement and thus stabilize the spine [15]. Nevertheless, in patients with LBP, the TrA and MF muscles are weak and delayed in activation, which ultimately leads to lumbar spine instability [8]. The faulty recruitment pattern of the TrA and MF causes poor support of the lumbar spine, increases the loading on the joints and ligaments of the lumbar spine, and ultimately leads to postural instability [8, 16].

Postural control is essential for daily life activities [17]. Posturography using center of pressure data examines integration and sensorimotor processing, which is a comprehensive method for evaluating motor control in CNSLBP patients [18]. A systematic review reported that postural stability was reduced in CNSLBP patients, indicating impaired motor control [18, 19]. Therefore, one of the crucial factors in restoring postural control in LBP patients is the co-contraction of TrA and MF muscles. Therapeutic strategies that focus on TrA and MF may improve postural stability in CNSLBP patients [17, 20]. Co-contraction of TrA and MF muscles may help with movement pattern restoration and lumbar-pelvic position sense [21].

European guidelines for the management of CNSLBP recommend exercise therapy as the primary treatment [22]. Core stability exercises (CSE) or motor control exercises have been considered as a treatment protocol for LBP [23]. The founders of CSE believe that LBP reduces the control and coordination of the trunk muscles, which affects movement control and spine stability [23]. As a result, CSE is intervention that aim to improve trunk muscle activation [22]. Using the principles of motor learning, CSE controls the intersegmental movement of the spine and improves the coordination of the spine and pelvis [22]. Previous studies have shown that CSE can reduce pain and disability and restore proprioception and postural stability in LBP patients [20, 24]. Another protocol, for trunk and spine stability is abdominal hallowing (AH) which focuses on the activation of local muscles, such as TrA, MF, and internal oblique (IO) through motor control [25]. Overall, the goal of AH is to minimize global muscle activity and increase local isolated muscle activation [25]. Therefore, AH can be effective in improving the activation pattern of the local muscles and increasing spine stability [25]. The goal of both approaches is to improve the patient's ability to control the trunk and spine stabilization during daily life activities. It seems that the CSE combined with AH provides stronger evidence about the effectiveness of these two protocols on postural stability in patients with CNSLBP. Therefore, this study was conducted to examine the effect of CSE combined with AH on postural balance in patients with CNSLBP. Based on the evidence, we hypothesized that CSE combined with AH has a significant effect on postural stability. It is hoped that the results of the study will provide new and practical insight for preventing and treating CNSLBP.

#### 2. Materials and Methods

#### **Research design**

This study is a double-blind (assessors/patients) randomized controlled trial that follows the Consort guidelines. Our study was performed between April to October 2021. The statistical population includes women with CNSLBP who were recruited from orthopedic or sports medicine clinics in Hamedan Province, Iran.

Patients were assigned to experimental and control groups. The interventions were performed for 8 weeks. Primary outcomes included pain, disability, proprioception, and secondary outcomes included static and dynamic balance.

## Sample size

We used G\*Power software, version 3.1.9.2 to estimate the sample size. According to Abbasi et al, the type I error was 0.05, and the type II error rate was planned to be 0.20 (statistical power 80%). The output of the software reported 26 subjects in a clinical trial. Considering a drop of 10%, we planned 30 patients in a clinical trial (15 patients for each group) [26].

#### **Participants**

The statistical population of this clinical trial was women with CNSLBP. Patients were recruited from August to November 2022 from orthopedic and sports medicine clinics in Hamedan Province. An orthopedic surgeon evaluated patients before recruitment. Thirty women with CNSLBP (age: 43.83±9.19 years, height: 156.97±5.84 cm, weight: 70.09±6.80 kg, body mass index (BMI): 28.58±3.91 kg/ m<sup>2</sup>, pain: 61±9.94) were recruited for the trial based on the diagnosis of the spine orthopedic physician. The inclusion criteria included visual analog scale ≥45, Oswestry disability index ≥25, no history of spine surgery, CNSLBP more than 12 weeks, and age range between 30-60 years. The exclusion criteria included pain in other segments of the body, and lower limb deformity (we used available tools, such as a caliper to evaluate the deformities of the lower limbs that affect low back pain. Caliper was used to evaluate foot pronation and knee alignment), history of spine surgery, use of painkillers in the past 6 months, physical therapy in the past year, participation in sports, sciatica pain (a spine orthopedic specialist separation sciatica pain from back pain through SLR and slump tests), spondylolysis, neuromuscular disorders, neurological, respiratory, muscle spasm, and vertebral fracture.

#### **Randomization and blinding**

We used random allocation software for randomization. Random codes were generated in blocks 4 and 6. We used sequentially numbered opaque sealed envelopes (SNOSE) to conceal the allocation. This study was planned as doubleblind (assessors/patient). The third researcher, who was not involved in data collection and was unaware of the research design, opened the envelopes and delivered them to the patients. Patients were unaware of group allocation. In addition, to maintain blinding, patients were instructed not to disclose information about the group allocation to the assessors. To ensure the effectiveness of blinding, the assessor was asked whether she/he was aware of the group allocation of the patients. If the assessors were aware of the group allocation of even one patient, the double-blind study was not considered.

## Instrument

We used the Biodex balance system device made by the Biodex Company in the United States to evaluate postural balance. Biodex has a movable circular platform that allows 20° of surface tilt in the anterior-posterior, and medial-lateral directions [27]. Biodex balance produces 3 indices of medial-lateral stability, anterior-posterior stability, and overall stability [27]. Previous studies reported ICC for the anterior-posterior index of 0.60-0.88, medial-lateral index of 0.64-0.94, and the overall stability index of 0.63-0.91 [27]. In addition, a universal plastic goniometer was used to evaluate lumbar proprioception.

## **Primary outcome**

#### Pain

We used a 100 mm visual analog scale for pain evaluation. This scale is a valid and reliable tool for pain evaluation in patients with LBP. On this scale, pain intensity is divided into four categories, 0-40 mm as no pain, 44-50 mm as mild pain, 45-74 mm as moderate pain, and 75-100 mm is classified as severe pain. The patient was asked to indicate the pain intensity during daily activities [28].

#### Disability

Oswestry disability index was used to disability evaluate (ICC=0.84) [29]. Disability is classified into four subcategories, including a score of less than 25 as the lowest level of disability, 25-50 as moderate disability, 50-75 as severe disability, and 75-100 as acute disability [28].

#### Proprioception

Newcomer's test was used to evaluate lumbar proprioception. The validity of the Newcomer test to evaluate lumbar proprioception is reported to be 87%. Three markers were placed on the acromion, iliac crest, and greater trochanter of the hip. The patients were taught to stand upright and spread both legs shoulder-width apart. The hands were placed crossed on the chest. Then, the goniometer's fixed arm was placed parallel to the femur, and the movable arm was placed in 30° trunk flexion position. The eyes were covered with a black cloth. Meanwhile, the patient was asked to move the trunk toward a 30° flexion position with a stable angular velocity. When the patient reached the 30° flexion position, audio feedback was provided. Meanwhile, the patients paused for 5 s to remember the target position and then returned to the initial position. To learn, they repeated the test 3 times. Then, the patient was asked to move the trunk to the  $30^{\circ}$  flexion position without feedback and to say "I arrived" when the patient target angle was reached [30]. This test was performed 3 times and the rest time between each trial was 5 s. Finally, the average of 3 trials was recorded as the position reconstruction error. If the average value of the errors was >3°, lumbar proprioception was considered healthy [31].

#### Secondary outcomes

#### Postural balance

The patients stood with bare feet on the platform and placed their hands crossed on chest. They were asked not to touch the fences during the trial and to stay stable when the platform was tilting. The test was performed with eyes closed; because patients with CNSLBP depend on the visual system for postural control [20]. In addition, the stability indices of the Biodex balance system are more reliable in the eyes-closed condition [27]. The Biodex balance platform stability ranges from level 1 as the most unstable to 12 as the least unstable [32]. The static and dynamic balance were evaluated in double-leg conditions. The stability level was set to "static" for static conditions and between "6 to 3" for dynamic conditions [28]. We only addressed the overall stability index [28]. The trial was conducted for 30 s with 3 repeated [27, 28]. The rest time between each trial was 10 s [28]. High values indicate poor posture stability.

## **Experimental protocol**

#### Core stability exercises

First, low levels of muscle contraction were performed as isometric and in positions with minimal loading. As the patient progressed, dynamic functional (activities that require spinal or limb movements) were planned by integrating local muscle co-contraction into functional tasks [23]. According to the previous study, CSE includes lying on the back, an abdominal brace with heel slide while lying, an abdominal brace while bridging, an abdominal brace while lying leg raises, an abdominal brace with bridging and raising the leg, an abdominal bracket in a vertical position with the abdominal, quadruped arm lift, leg lift, hand, and leg lift alternated [7]. CSE was performed 3 times a week, each session lasting 40 minutes for 8 weeks. Core stability exercises were performed in two stages. First, the muscle endurance factor was considered (weeks 1-4). A total of 12-15 repetitions and 4-7 sets are very effective for local muscle

endurance [33, 34]. The rest time depends on exercise goals [34]. Minimizing recovery between sets is a crucial stimulant for improving local muscular endurance [34]. Likewise, we included 30 s of rest to increase muscular endurance [35]. Progressive overload is essential for maximal muscle fiber recruitment [34]. In this line, the progressive principle was maintained as the position for 10 s in each repetition [35].

In the second stage, increased strength and muscle activation were considered (weeks 5-8). Ten repetitions and 1-3 sets are effective for increasing local muscle strength [34]. However, the studies have shown that 3 sets are more effective in increasing muscle strength [34]. Likewise, we considered 3 sets to increase strength. It is recommended that when the repetitions are considered to be 10 repetitions, the rest period to increase muscle strength is considered to be 1 minute [34]. In addition, it has been reported that slow-twitch muscle fibers recover very quickly and almost completely within 1 minute [36]. Consequently, we rested for 1 minute between each set. The progressive principle in this stage included changing the exercise position, resistance through the band, and slow to moderate execution velocity of the exercises.

#### Abdominal hallowing

Target muscles in AH included TrA, IO (low fibers), and MF (deep fibers) [37]. The co-contraction of trunk local muscles should be isolated from global muscles, such as rectus abdominis and external abdominal oblique [37]. AH in the first phase (weeks 1-4) focused on lowload isolated activation of TrA and MF muscles [11]. Meanwhile, for the first week, the wall support standing positions were considered for the TrA isolated activation [37]. The patients were asked to stand with their back against the wall with hips in flexion and extended knees. The distance from the wall to the heel was 6 inches. The patients were asked to pull the lower anterior abdominal

wall "up and in" towards the spine. MF isolation was activated by arm raising on the opposite side during AH. The second phase (weeks 5-8) emphasized the accuracy and duration of AH [11]. The co-contraction of TrA and MF muscles was performed by controlling upper and lower limb movements and progressed to high-loading positions in lying and sitting positions. During AH, patients were instructed to monitor the contraction of TrA and MF by palpating the muscles. To TrA contraction, the index and middle fingers touched the 2 cm area medial to the anterior superior iliac spines. In contrast, for MF contraction, the index and middle fingers were placed near the L5 spinous process [11]. During AH, no movement was performed in the spine, ribs, or pelvis [37]. The patients were taught to hold the abdominal contraction for 10 s and breathe continuously [37]. CSE combined with AH was performed 3 times a week, each session was 20 minutes for 8 weeks.

#### Statistical analysis

SPSS software, version 26 was used for statistical analysis. The Shapiro-Wilk test was used to ensure the data's normal distribution. The equality of error variances was checked using Levene's test. The covariance test was used for between-group comparison. The effect size was calculated through Cohen's index. Cohen's index is classified into three categories (0.01-0.059 small effect size, 0.06-0.14 medium effect size, and 0.14 large effect size) [38]. A significance level of P<0.05 was considered.

#### **3. Results**

Table 1 presents the demographic characteristics of the groups. The results of the Shapiro-Wilk test showed no statistically significant difference between the groups for demographic characteristics. Therefore, the data distribution is normal and the variances are equal. Table 2 presents the results of the covariance test. Covariance re-

Table 1. Shapiro-Wilk test result (n=	15)
---------------------------------------	-----

Variables	Mea	Р	
variables	Experimental	Control	— P
Age (y)	45.66±8.83	42.00±9.48	0.154
Height (cm)	154.78±5.41	159.16±5.58	0.856
Weight (kg)	69.74±6.76	70.45±7.07	0.221
BMI (kg/m²)	29.34±3.34	27.81±4.38	0.304
Pain (VAS)	62.33±10.15	59.33±10.49	0.154
AS: Visual analogue scale.			PHYSICAL TREAT MENT

VAS: Visual analogue scale.

Variables	Groups –	Mean±SD		95% CI (Post-test)		F		Р
		Pre	Post	Lower	Upper	•	η²	r
Pain (VAS)	Experimental	62.33±10.15	29.00±12.84	21.88	36.11	6.041	0.183	0.021*
	Control	59.66±9.90	53.20±6.66	32.89	73.50			
Disability (numeric)	Experimental	44.86± 11.12	26.13±11.52	19.75	32.51	8.368	0.237	0.007*
	Control	40.53±12.50	34.93±14.28	27.02	42.84			
Proprioception (angle)	Experimental	7.46±3.79	2.14±3.16	0.39	3.89	35,232	0.566	0.000*
	Control	7.40±6.50	9.38±3.48	7.45	11.30	55.252	0.566	0.000
Static OS (degrees)	Experimental	1.86±0.46	1.11±0.53	0.82	1.41	15.660	0.367	0.000*
	Control	1.47±0.87	1.79±0.78	1.45	2.32			0.000
Dynamic OS (degrees)	Experimental	3.48±0.91	1.90±0.51	1.61	2.18	83.980	0.757	0.000*
	Control	2.92±0.62	3.80±0.56	3.49	4.12		0.757	0.000

Table 2. ANCOVA results and descriptive statistics for outcome measuring

VAS: Visual analogue scale; OS: Overall stability.

\*Significant difference between groups (P<0.05)

sults showed that CSE combined with AH had a significant difference with a high effect size in reducing pain (P=0.021,  $\eta p^2$ =0.183), disability (P=0.007,  $\eta p^2$ =0.237) and position reconstruction error (P=0.000,  $\eta p^2$ =0.566). In addition, the covariance results showed that CSE combined with AH had a significant difference with high effect size in improving static (P=0.008,  $\eta p^2$ =0.367) and dynamic balance (P=0.008,  $\eta p^2$ =0.757).

#### 4. Discussion

This study was conducted to examine the effect of CSE combined with AH on postural balance in patients with CNSLBP. The primary outcome showed that CSE combined with AH had a significant effect on reducing pain, disability, and position reconstruction error. Also, the secondary outcome showed that CSE combined with AH had a significant effect on improving postural balance. The results were consistent with some of the studies focusing on CSE and AH. For example, Ogunniran et al. reported pain reduction [7], Kim et al. reported disability reduction [39], Hlaing et al. (2021) reported proprioception restoration [11], and Puntumetakul et al. (2021) reported improved postural stability [40]. Panjabi stated that spinal instability is likely due to dysfunction of the passive (non-contractile) or active components (trunk muscles) of the spine or a defect in neural control, leading to LBP [17]. Stabilizing exercises facilitate local muscle activity and inhibit global muscle activity [7, 41]. This mechanism increases stability and reduces the overPHYSICAL TREAT MENTS

load of the lumbar spine; as a result, pain and disability decrease [11, 21].

Pain harms the sensory feedback from muscle spindle afferents. The proprioceptive deficiency causes feedforward and feedback motor control disorders and inefficiency of the sensory-motor system [11, 21]. As a result, joint stability decreases and posture oscillation increases. Therefore, poor stability in LBP patients may be due to altered sensory feedback from the lumbar spine [42]. This pathophysiology may be due to a defect in the central processing of sensory information from mechanoreceptors [42]. The sensory impulse defect can lead to greater movements of the lumbar spine; as a result, the body sway responses increase for postural control [43]. Therefore, faulty motor planning caused by pain disrupts feedforward and feedback control [21]. The TrA and MF muscles are the main sources of trunk sensory inputs and play a critical role in providing sensory impulses to the posture control system [11, 20]. Lee reported that AH focuses on the deep muscle system and is more effective than CSE for postural stability [44]. Recruitment at the low threshold of TrA and MF muscles can proprioception facilitation by motor integration [11].

The body core consists of passive (thoracolumbar fascia) and active components (trunk muscles) [45]. The thoracolumbar fascia acts as a "nature's back belt" [46]. TrA and MF muscles act as primary generators of intraabdominal pressure [15]. Of The total core muscles, TrA has the widest and closest connection with the thoracolumbar fascia and is essential for spinal stability [46]. Due to the horizontal orientation of the TrA muscle, the contraction of this muscle can increase the tension in the thoracolumbar fascia and thus increase intra-abdominal pressure [28, 41]. Increased intra-abdominal pressure leads to the stiffness of stabilizing muscles and provides spine segmental stability [15, 41, 47]. In addition, previous studies showed that AH increases the isolated activity of the TrA muscle and the lower fibers of the IO muscle [37, 41]. The TrA and IO muscles act synergistically [37]. TrA and IO muscles can strengthen the lumbar-pelvic complex and improve force transmission from the upper limb to the lower limb. Meanwhile, if core stability is poor, a higher force is imposed on the spine, leading to fatigue and risk of injury [15]. Another effective factor in the balance is the interaction between the feedforward and feedback mechanisms [45]. TrA and MF muscles are activated before movements and can increase postural stability [15]. Evidence shows that feedforward control can be restored with exercises focused on core muscle recruitment [20]. Meanwhile, it has been reported that the feedforward mechanism and activation of trunk muscles in the motor cortex are restored following stabilization exercise emphasizing the co-contraction of TrA and MF muscles [11, 20].

On the other hand, McNeill reported that CSE affects the brain and sensory-motor inputs [48]. Modulation of the central nervous system is vital for motor integration and may strengthen joint position sense [21]. With the co-contraction of TrA and MF muscles, the thoracolumbar fascia is activated as a proprioceptor and provides feedback [46]. Increasing the sensory impulse about joint position and changes in the length and tension of muscles improves the ability of the nervous system to recruit muscles, increasing the number of movement units, and muscle coordination [10]. Following proprioceptive facilitation, the feed-forward and feedback motor control is improved; as a result, it leads to joint stability, coordination, and postural stability [10, 49]. Finally, sensorymotor integration following CSE increases the ability of patients to maintain the center of pressure on a supportive basis [7]. CSE combined with AH is based on motor learning, which, by focusing on the TrA and MF muscles recruitment, can reverse the pain experience in the motor cortex, and modify faulty motor integration and muscle imbalance [7, 11].

In this study, we had three limitations. First, our sample size was only women, second, the unavailability of ultrasound and electromyography for assessment of the thickness and activity of the core muscles, third, failure to follow up results in the long term. Therefore, we recommended that future studies evaluate the effect of CSE combined with AH in men and women on the thickness and activation of the core muscles in the long term.

## 5. Conclusion

The CSE combined with AH affects the active, passive, and nervous systems. It is believed that stability exercises can improve the activation pattern in local muscles, facilitate proprioceptive, control spine movement, and finally lead to postural stability. Therefore, following the functional improvement of the three subgroups of active, passive, and nervous after CSE combined with AH, lumbar spine stability and postural stability are improved.

## **Ethical Considerations**

#### Compliance with ethical guidelines

The Ethics Committee of Bio-Medical Research of Bu-Ali Sina University approved the present study (Code: IR.BASU.REC.1399.036). This research followed the Helsinki Declaration of 2008.

## Funding

This research did not receive any specific grant from funding agencies in the public, commercial, or not-forprofit sectors.

#### Authors' contributions

All authors contributed equally to preparing this article.

#### **Conflict of interest**

The authors declared no conflict of interest.

#### Acknowledgments

The authors thank all patients for participating in this clinical trial.

#### References

Turci AM, Nogueira CG, Carrer HC, Chaves TC. Self-administered stretching exercises are as effective as motor control exercises for people with chronic non-specific low back pain: A randomised trial. Journal of Physiotherapy. 2023; 69(2):93-9. [DOI:10.1016/j.jphys.2023.02.016] [PMID]

- [2] Hoy D, Bain C, Williams G, March L, Brooks P, Blyth F, et al. A systematic review of the global prevalence of low back pain. Arthritis and Rheumatism. 2012; 64(6):2028-37. [DOI:10.1002/art.34347] [PMID]
- [3] Farahpour N, Jafarnezhadgero A, Allard P, Majlesi M. Muscle activity and kinetics of lower limbs during walking in pronated feet individuals with and without low back pain. Journal of Electromyography and Kinesiology. 2018; 39:35-41. [DOI:10.1016/j.jelekin.2018.01.006] [PMID]
- [4] Abd Rahman NA, Li S, Schmid S, Shaharudin S. Biomechanical factors associated with non-specific low back pain in adults: A systematic review. Physical Therapy in Sport. 2023; 59:60-72. [DOI:10.1016/j.ptsp.2022.11.011] [PMID]
- [5] Goossens N, Janssens L, Caeyenberghs K, Albouy G, Brumagne S. Differences in brain processing of proprioception related to postural control in patients with recurrent nonspecific low back pain and healthy controls. NeuroImage: Clinical. 2019; 23:101881. [DOI:10.1016/j.nicl.2019.101881] [PMID] [PMCID]
- [6] Verbrugghe J, Agten A, Stevens S, Vandenabeele F, Roussel N, Verbunt J, et al. High intensity training improves symptoms of central sensitization at six-month follow-up in persons with chronic nonspecific low back pain: Secondary analysis of a randomized controlled trial. Brazilian Journal of Physical Therapy. 2023; 27(2):100496. [DOI:10.1016/j. bjpt.2023.100496] [PMID] [PMCID]
- [7] Ogunniran IA, Akodu AK, Odebiyi DO. Effects of kinesiology taping and core stability exercise on clinical variables in patients with non-specific chronic low back pain: A randomized controlled trial. Journal of Bodywork and Movement Therapies. 2023; 33:20-7. [DOI:10.1016/j.jbmt.2022.09.013] [PMID]
- [8] Lopes S, Correia C, Félix G, Lopes M, Cruz A, Ribeiro F. Immediate effects of pilates based therapeutic exercise on postural control of young individuals with non-specific low back pain: A randomized controlled trial. Complementary Therapies in Medicine. 2017; 34:104-10. [DOI:10.1016/j. ctim.2017.08.006] [PMID]
- [9] Wang X, Zheng J, Yu Z, Bi X, Lou S, Liu J, et al. A metaanalysis of core stability exercise versus general exercise for chronic low back pain. PLoS One. 2012; 7(12):e52082. [DOI:10.1371/journal.pone.0052082] [PMID] [PMCID]
- [10] Ahmadi M, Yalfani A, Gandomi F. [Effect of twelve weeks of sensorimotor training on pain, improvement proprioception, muscle strength, and postural control in men with patellofemoral pain syndrome: A randomized single-blind clinical trial (Persian)]. The Scientific Journal of Rehabilitation Medicine. 2021; 10(1):1-3. [DOI:10.22037/ jrm.2020.112859.2271]
- [11] Hlaing SS, Puntumetakul R, Khine EE, Boucaut R. Effects of core stabilization exercise and strengthening exercise on proprioception, balance, muscle thickness and pain related outcomes in patients with subacute nonspecific low back pain: A randomized controlled trial. BMC Musculoskeletal Disorders. 2021; 22(1):998. [DOI:10.1186/s12891-021-04858-6] [PMID] [PMCID]
- [12] Wanpen S, Boucaut R. Balance control in patients with subacute non-specific low back pain, with and without lumbar instability : A cross-sectional study. Journal of Pain Research. 2020; 795-803. [DOI:10.2147/JPR.S232080] [PMID] [PMCID]

- [13] Sipko T, Glibowski E, Kuczyński M. Acute effects of proprioceptive neuromuscular facilitation exercises on the postural strategy in patients with chronic low back pain. Complementary Therapies in Clinical Practice. 2021; 44:101439. [DOI:10.1016/j.ctcp.2021.101439] [PMID]
- [14] Matthijs OC, Dedrick GS, James CR, Brismée JM, Hooper TL, McGalliard MK, et al. Co-contractive activation of the superficial multifidus during volitional preemptive abdominal contraction. PM & R. 2014; 6(1):13-21. [DOI:10.1016/j. pmrj.2013.08.606] [PMID]
- [15] Coulombe BJ, Games KE, Neil ER, Eberman LE. Core stability exercise versus general exercise for chronic low back pain. Journal of Athletic Training. 2017; 52(1):71-2. [DOI:10.4085/1062-6050-51.11.16] [PMID] [PMCID]
- [16] Frizziero A, Pellizzon G, Vittadini F, Bigliardi D, Costantino C. Efficacy of core stability in non-specific chronic low back pain. Journal of Functional Morphology and Kinesiology. 2021; 6(2):37. [DOI:10.3390/jfmk6020037] [PMID] [PMCID]
- [17] Wang H, Zheng J, Fan Z, Luo Z, Wu Y, Cheng X, et al. Impaired static postural control correlates to the contraction ability of trunk muscle in young adults with chronic non-specific low back pain: A cross-sectional study. Gait & Posture. 2022; 92:44-50. [DOI:10.1016/j.gaitpost.2021.11.021] [PMID]
- [18] Sung W, Abraham M, Plastaras C, Silfies SP. Trunk motor control deficits in acute and subacute low back pain are not associated with pain or fear of movement. The Spine Journal. 2015; 15(8):1772-82. [DOI:10.1016/j.spinee.2015.04.010] [PMID] [PMCID]
- [19] Mazaheri M, Coenen P, Parnianpour M, Kiers H, van Dieën JH. Low back pain and postural sway during quiet standing with and without sensory manipulation: A systematic review. Gait & Posture. 2013; 37(1):12-22. [DOI:10.1016/j. gaitpost.2012.06.013] [PMID]
- [20] Salavati M, Akhbari B, Takamjani IE, Bagheri H, Ezzati K, Kahlaee AH. Effect of spinal stabilization exercise on dynamic postural control and visual dependency in subjects with chronic non-specific low back pain. Journal of Bodywork and Movement Therapies. 2016; 20(2):441-8. [DOI:10.1016/j. jbmt.2015.10.003] [PMID]
- [21] Puntumetakul R, Chalermsan R, Hlaing SS, Tapanya W, Saiklang P, Boucaut R. The effect of core stabilization exercise on lumbar joint position sense in patients with subacute non-specific low back pain: A randomized controlled trial. Journal of Physical Therapy Science. 2018; 30(11):1390-5. [DOI:10.1589/jpts.30.1390] [PMID] [PMCID]
- [22] Gomes-Neto M, Lopes JM, Conceição CS, Araujo A, Brasileiro A, Sousa C, et al. Stabilization exercise compared to general exercises or manual therapy for the management of low back pain: A systematic review and meta-analysis. Physical Therapy in Sport. 2017; 23:136-42. [DOI:10.1016/j. ptsp.2016.08.004] [PMID]
- [23] Shamsi MB, Rezaei M, Zamanlou M, Sadeghi M, Pourahmadi MR. Does core stability exercise improve lumbopelvic stability (through endurance tests) more than general exercise in chronic low back pain? A quasi-randomized controlled trial. Physiotherapy Theory and Practice. 2016; 32(3):171-8. [DOI:10.3109/09593985.2015.1117550] [PMID]

- [24] Kim TH, Kim EH, Cho HY. The effects of the CORE programme on pain at rest, movement-induced and secondary pain, active range of motion, and proprioception in female office workers with chronic low back pain: A randomized controlled trial. Clinical Rehabilitation. 2015; 29(7):653-62. [DOI:10.1177/0269215514552075] [PMID]
- [25] Kahlaee AH, Ghamkhar L, Arab AM. Effect of the abdominal hollowing and bracing maneuvers on activity pattern of the lumbopelvic muscles during prone hip extension in subjects with or without chronic low back pain: A preliminary study. Journal of Manipulative and Physiological Therapeutics. 2017; 40(2):106-17. [DOI:10.1016/j.jmpt.2016.10.009] [PMID]
- [26] Abbasi S, Hadian Rasanani MR, Ghotbi N, Olyaei GR, Bozorgmehr A, Rasouli O. Short-term effect of kinesiology taping on pain, functional disability and lumbar proprioception in individuals with nonspecific chronic low back pain: A double-blinded, randomized trial. Chiropractic & Manual Therapies. 2020; 28(1):63. [DOI:10.1186/s12998-020-00349-y] [PMID] [PMCID]
- [27] Sherafat S, Salavati M, Takamjani IE, Akhbari B, Mohammadirad S, Mazaheri M, et al. Intrasession and intersession reliability of postural control in participants with and without nonspecific low back pain using the Biodex Balance System. Journal of Manipulative and Physiological Therapeutics. 2013; 36(2):111-8. [DOI:10.1016/j.jmpt.2012.12.005] [PMID]
- [28] Yalfani A, Raeisi Z, Koumasian Z. Effects of eight-week water versus mat pilates on female patients with chronic nonspecific low back pain: Double-blind randomized clinical trial. Journal of Bodywork and Movement Therapies. 2020; 24(4):70-5. [DOI:10.1016/j.jbmt.2020.06.002] [PMID]
- [29] Inani SB, Selkar SP. Effect of core stabilization exercises versus conventional exercises on pain and functional status in patients with non-specific low back pain: A randomized clinical trial. Journal of Back and Musculoskeletal Rehabilitation. 2013; 26(1):37-43. [DOI:10.3233/BMR-2012-0348] [PMID]
- [30] Ahmadi M, Yalfani A, Gandomi F, Rashid K. The effect of twelve-week neurofeedback training on pain, proprioception, strength and postural balance in men with patellofemoral pain syndrome: A double-blind randomized control trial. Journal of Rehabilitation Sciences & Research. 2020; 7(2):66-74. [DOI:10.30476/jrsr.2020.84868.1067]
- [31] Letafatkar A, Nazarzadeh M, Hadadnezhad M, Farivar N. The efficacy of a HUBER exercise system mediated sensorimotor training protocol on proprioceptive system, lumbar movement control and quality of life in patients with chronic non-specific low back pain. Journal of Back and Musculoskeletal Rehabilitation. 2017; 30:767-78. [DOI:10.3233/BMR-150404] [PMID]
- [32] Ehsani F, Arab AM, Jaberzadeh S, Salavati M. Ultrasound measurement of deep and superficial abdominal muscles thickness during standing postural tasks in participants with and without chronic low back pain. Manual Therapy. 2016; 23:98-105. [DOI:10.1016/j.math.2016.01.003] [PMID]
- [33] Anemaet WK, Hammerich AS. A framework for exercise prescription. Topics in Geriatric Rehabilitation. 2014; 30(2):79-101. [DOI:10.1097/TGR.00000000000011]

- [34] Kraemer WJ, Ratamess NA. Fundamentals of resistance training: Progression and exercise prescription. Medicine & Science in Sports & Exercise. 2004; 36(4):674-88. [DOI:10.1249/01.MSS.0000121945.36635.61] [PMID]
- [35] Schilling JF, Murphy JC, Bonney JR, Thich JL. Effect of core strength and endurance training on performance in college students: Randomized pilot study. Journal of Bodywork and Movement Therapies. 2013; 17(3):278-90. [DOI:10.1016/j. jbmt.2012.08.008] [PMID]
- [36] Wallden M. Designing effective corrective exercise programs: The importance of dosage. Journal of Bodywork and Movement Therapies. 2015; 19(2):352-6. [DOI:10.1016/j. jbmt.2015.02.006] [PMID]
- [37] Chanthapetch P, Kanlayanaphotporn R, Gaogasigam C, Chiradejnant A. Abdominal muscle activity during abdominal hollowing in four starting positions. Manual Therapy. 2009; 14(6):642-6. [DOI:10.1016/j.math.2008.12.009] [PMID]
- [38] Yalfani A, Ahmadi M, Asgarpoor A, Ahmadi AH. Effect of foot orthoses on dynamic balance in taekwondo athletes with flexible flatfoot: A randomized controlled trial. The Foot. 2023; 56:102042. [DOI:10.1016/j.foot.2023.102042] [PMID]
- [39] Kim M, Kim M, Oh S, Yoon B. The effectiveness of hollowing and bracing strategies with lumbar stabilization exercise in older adult women with nonspecific low back pain: A quasi-experimental study on a community-based rehabilitation. Journal of Manipulative and Physiological Therapeutics. 2018; 41(1):1-9. [DOI:10.1016/j.jmpt.2017.06.012] [PMID]
- [40] Puntumetakul R, Saiklang P, Yodchaisarn W, Hunsawong T, Ruangsri J. Effects of core stabilization exercise versus general trunk-strengthening exercise on balance performance, pain intensity and trunk muscle activity patterns in clinical lumbar instability patients: A single blind randomized trial. Walailak Journal of Science and Technology. 2021; 18(7):1-13. [DOI:10.48048/wjst.2021.9054]
- [41] Suehiro T, Mizutani M, Watanabe S, Ishida H, Kobara K, Osaka H. Comparison of spine motion and trunk muscle activity between abdominal hollowing and abdominal bracing maneuvers during prone hip extension. Journal of Bodywork and Movement Therapies. 2014; 18(3):482-8. [DOI:10.1016/j.jbmt.2014.04.012] [PMID]
- [42] Behennah J, Conway R, Fisher J, Osborne N, Steele J. The relationship between balance performance, lumbar extension strength, trunk extension endurance, and pain in participants with chronic low back pain, and those without. Clinical Biomechanics. 2018; 53:22-30. [DOI:10.1016/j.clinbiomech.2018.01.023] [PMID]
- [43] da Silva RA, Vieira ER, Léonard G, Beaulieu LD, Ngomo S, Nowotny AH, et al. Age-and low back pain-related differences in trunk muscle activation during one-legged stance balance task. Gait & Posture. 2019; 69:25-30. [DOI:10.1016/j. gaitpost.2019.01.016] [PMID]
- [44] Lee JY, Lee DY. The effect of therapeutic abdominal drawing-in maneuver using ultrasonography on lateral abdominal muscle thickness and balance. Journal of Back and Musculoskeletal Rehabilitation. 2018; 31(6):1139-43. [DOI:10.3233/BMR-171080] [PMID]
- [45] Borghuis J, Hof AL. The Importance of sensory-motor control in providing core stability. Sport Med. 2008; 38(11):893-916. [DOI:10.2165/00007256-200838110-00002] [PMID]

- [46] Arendt EA. Core strengthening. Instructional course lectures. 2007; 56:379-84. [PMID]
- [47] Shimizu I, Tateuchi H, Motomura Y, Morishita K, Masaki M, Ichihashi N. Abdominal girth as an index of muscle tension during abdominal hollowing: Selecting the optimal training intensity for the transversus abdominis muscle. Journal of Biomechanics. 2019; 89:72-7. [DOI:10.1016/j.jbiomech.2019.04.018] [PMID]
- [48] McNeill W. Core stability is a subset of motor control. Journal of Bodywork and Movement Therapies. 2010; 14(1):80-3. [DOI:10.1016/j.jbmt.2009.10.001] [PMID]
- [49] Yalfani A, Ahmadi M, Gandomi F. [The effect of twelve weeks of sensorimotor exercises on distribution plantar pressure variables and symmetry index in patients with patellofemoral pain syndrome: A randomized double-blind clinical trial (Persian)]. Studies in Medical Sciences. 2020; 31(6):445-58. [Link]